Author's response to reviews

Title: Cardiac conduction abnormalities and congenital immunodeficiency in a child with Kabuki syndrome: case report.

Authors:
- Maulik R Shah (shahmr@slu.edu)
- Brian Bogucki (luckybogucki@aol.com)
- Melissa Mavers (maversmm@SLU.EDU)
- Daphne E deMello (demellde@slu.edu)
- Alan Knutsen (knutsenm@slu.edu)

Version: 3 Date: 29 June 2005

Author's response to reviews: see over
June 29, 2005

Assistant Editor
BMC Medical Genetics

Dear Alison Shaw:

We are pleased to have revised the paper according to the final comments from the reviewers and submitted it for publication. We have detailed the changes below.

Formatting – The following formatting changes were made according to the editorial staff.

1. Affiliations were written in full according to the format given with superscript links to each author.

2. The email and contact information for the corresponding author is given on the title page.

3. The keyword section was removed.

4. The competing interests section was modified to declare that the authors have no competing interests.

5. Section headings and the title now use sentence case.

6. Manuscript sections are in the prescribed format.

Reviewers Concerns – Neither of the reviewers had a significant corrections to note. The second reviewer suggested consistency in labeling the syndrome throughout the paper and this has been corrected.

Sincerely,

Maulik R. Shah M.D. Ph.D.
Instructor
Division of Medical Genetics

Cardinal Glennon Children’s Hospital
Saint Louis University